Author's response to reviews

Title: Anthropometry measures and prevalence of obesity in the urban adult population of Cameroon: an update from the Cameroon Burden of Diabetes Baseline Survey

Authors:

Raoul M Kamadjieu (raoul.kamadjieu@gmail.com)
Richard Edwards (richard.edwards@otago.ac.nz)
Joseph S Atanga (shujoe@yahoo.co.uk)
Emmanuel C Kiawi (kiawi_emma@yahoo.com)
Nigel Unwin (unwinn@who.int)
Jean-Claude Mbanya (jcmbanya@yahoo.co.uk)

Version: 5 Date: 12 September 2006

Author's response to reviews:

The requested correction were made